page 2

2.3 Artificial Neural Networks

• Haykin, S. (1994) Neural Networks: A Comprehensive Foundation. New York: Macmillan. • Hertz, J., Krogh, A., and Palmer, R. G. (1991) Introduction to the Theory of Neural Computation.

Addison-Wesley.

2.4 Connectionism

• Rumelhart, D. E. and McClelland, J. L. (1986) Parallel Distributed Pr ocessing: Explorations in the Microstruction of Cognition. MIT Press.

2.5 Philosophy

• Churchland, P. S. (1986) Neurophilosophy: Toward a Unified Understanding of the Mind/Brain.

MIT Press. • Churchland, P. M. 1989) A Neurocomputational Perspective: The Nature of Mind and the Structure of Science. MIT Press.

Useful Terms and General Refer ences

Computational Models of Neural Systems 15-880(B) Spring ’95 Version of 25 January 1995 19:59

David S. Touretzky

1. Anatomical T erms

rostral vs. caudal rostral = ‘‘toward the snout’’; caudal = ‘‘toward the tail’’

lateral vs. medial lateral = ‘‘toward the sides/edges’’; medial = ‘‘toward the middle’’

pyramids pyramidal cells, so named because the cell body has a pyramidal shape

apical vs. basal dendrites dendrites originating at the tip vs. the base of the pyramid

afferent vs. efferent fibers input vs. output, or sensory vs. motor fibers

collaterals fibers running along side each other

recurrent feeding back on itself

layers I-VI layers of the cortex, each with a distinct connection pattern

For additional definitions, see the glossary in The Computational Brain.

2. General Refer ences

2.1 Neurophysiology

• Kandel, E. R., Schwartz, J. H., and Jessell, T. M. (1991) Principles of Neural Science, third edition. New York: Elsevier.

• Nicholls, J. G., Martin, A. R., and Wallace, B. G. (1992) From Neuron to Brain, third edition. Sunderland, MA: Sinauer Associates.

• Shepherd, G. M. (ed.) (1990) The Synaptic Organization of the Brain, third edition. Oxford University Press.

2.2 Computational and Systems Neur oscience

• Churchland, P. S. and Sejnowski, T. J. (1992) The Computational Brain. MIT Press. • Crick, F. H. C. (1994) The Astonishing Hypothesis: The Scientific Search for the Soul. New York: Charles Scribner’s Sons.

• Gluck, M. A. and Rumelhart, D. E. (1990) Neuroscience and Connectionist Theory. Hillsdale, NJ:

Lawrence Erlbaum Associates.